Author's response to reviews

Title: Toxoplasma, Toxocara and Tuberculosis co-infection in a four year old child

Authors:

Randeewari Guneratne (randeewari@yahoo.com)
Devan Mendis (devanmendis@gmail.com)
Tharaka Bandara (tharakabandara@hotmail.com)
Sumadhya D Fernando (fernddeep@gmail.com)

Version: 4 Date: 24 May 2011

Author's response to reviews: see over
Dear Sir/Madam

We have attended to the revisions given below. I would like to clarify the point Additional Files. I assume this refers to tables or figures, neither of which I will be uploading. Excluding this fact all changes have been done.

Thank You

Deepika

Major revisions (we require the you to make these changes)
1) Author list: The author list in the manuscript does not match the submission system. For example in the submission system author 1 reads: Randeewari Guneratne but in the manuscript it reads Randeewari Chathurika Wijesinghe Guneratne. Please can you amend either the submission system or manuscript so the two match.

This has been done

Abstract: Please ensure the abstract is in the submitted manuscript just after the title page and before the main body of the text- currently there is no abstract section in the submitted manuscript. In addition the abstract in the submission system and in the manuscript should adhere to the following structure: Background, Case Presentation, Conclusions.

This has been done.

References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example ‘text [1].’

This has been done

Additional files: Please note the additional files are not cited in the main text which is a requirement for them to appear in the final version. In addition it was noticed that you have included a letter to the editor and comments made by the editorial board as additional files which is not necessary so please remove the additional files if they are not directly related to the manuscript.
Please clarify what the additional files are. Since this is a case report there are no tables or figures and the main text does not refer to any other additional documents.

I have removed the letter and comments to the editorial board from the main text.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)
Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

All highlights have been removed

Abbreviations: Please format these in to a paragraph.

Done

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Done

Thank you